Author's response to reviews

Title: Cost-effectiveness of collaborative care including PST and an antidepressant treatment algorithm for the treatment of major depressive disorder in primary care; a randomised clinical trial.

Authors:

Marjoliek A IJff (mijff@trimbos.nl)
Klaas ML Huijbregts (khuijbregts@trimbos.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Aartjan TF Beekman (aartjanb@ggzba.nl)
Leona Hakkaart-van Roijen (l.hakkaart-vanroijen@erasmusmc.nl)
Frans Rutten (f.rutten@erasmusmc.nl)
Jurgen Unutzer (unutzer@u.washington.edu)
Christina M van der Feltz-Cornelis (cfeltz@trimbos.nl)

Version: 5 Date: 28 February 2007

Author's response to reviews: see over
Dear editor,

I am delighted to learn that BMC Health Services Research has decided to publish our manuscript pending some format changes! Hence I have resubmitted our manuscript ‘Cost-effectiveness of Collaborative Care including PST and an Antidepressant Treatment Algorithm for the treatment of Major Depressive Disorder in primary care; a Randomised Clinical Trial’ (Manuscript ID 1945482116126357) after making the requested format changes (i.e. sentence case in the title, removing the term et al. in the reference list and adding all authors, removing the figure title in the accompanying figure, cropping the figure and finally some small textual changes).

If I can provide you with any other information, please do not hesitate to contact me.

Best wishes,

Marjoliek IJff